Author's response to reviews

Title: Risk of cancer among HIV-infected patients from a population-based retrospective cohort study: implications for cancer prevention

Authors:

Chang-Hua Chen (changhua@cch.org.tw)
Chih-Yuan Chung (45413@cch.org.tw)
Li-Hsuan Wnag (shiuan@tmu.edu.tw)
Che Lin (chelin1chelin1@yahoo.com.tw)
Hsiu-Li Lin (linali@ms1.hinet.net)
Hsiu-Chen Lin (jane2@tmu.edu.tw)

Version: 2 Date: 1 May 2014

Author’s response to reviews:

Dear editor:

I had described the sentence about "ethic" in the manuscript, and added the line numbering and e-mail address of all authors.